Author’s response to reviews

Title: Diagnostic accuracy of optical coherence tomography angiography for choroidal neovascularization: a systematic review and meta-analysis

Authors:
Rui Wang (1845584559@qq.com)
Zhenzhen Liang (liangzz15@mails.jlu.edu.cn)
Xin Liu (xiaorui0110@163.com)

Version: 2 Date: 02 Jul 2019

Author’s response to reviews:

Corrections

Author(s): Rui Wang, M.D., Zhenzhen Liang, Ph.D., Xin Liu, Ph.D.

Article title: Diagnostic accuracy of optical coherence tomography angiography for choroidal neovascularization: a systematic review and meta-analysis.

Manuscript Number: BOPH-D-19-00342

Dear Editor:

Thank you for your comments and suggestions. We have revised accordingly, and corrections were listed below point by point.

Editors’ comments:
Q1: Copyedit

Your manuscript needs to be copyedited, for example, character format. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service.

A1: Based on your advice, we asked a native English speaking colleague to copyedit our manuscript.
Q2: Section heading

Please rename the section heading 'Introduction' to 'Background', 'Materials and methods' to 'Methods' in the main text of this manuscript.

A2: According to your advice, we changed section heading 'Introduction' into 'Background' (on page 4, line 8), 'Materials and methods' to 'Methods' in the main text of this manuscript (on page 6, line 1).

Q3: Authors’ contributions

Please also ensure that all authors have read and approve of the final version and add a statement confirming this within this section.

A3: All authors read and approved the final version and we wrote a statement confirming this within this section (on page 17, line 6).

Q4: Funding

The role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript should be declared within this section.

A4: We declared the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript (on page 16, line 19 to page 17, line 1).

Q5: Clean copy

On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

A5: We removed tracked changes and highlighting and included only a single clean copy of the manuscript.